$5$ conference papers: CALCO 2025~\cite{aristoteActiveLearningUpwardClosed2025};
CSL 2024~\cite{aristoteActiveLearningDeterministic2024}; FoSSaCS
2026~\cite{aristoteLearningBottomUpTree2026}; LICS
2025~\cite{aristoteMonotoneWeakDistributive2025}; STACS
2025~\cite{aristoteLearningWeightedAutomata2025}.

\smallskip

$>24$ talks given. $5$ invited talks: Oxford University, Université
Aix-Marseille ($\times 2$), Université de Rouen, Université Savoie-Mont-Blanc

\smallskip

\textbf{Awards:}

\smallskip

\cventry{ETAPS 2026~\cite{aristoteLearningBottomUpTree2026}}{Distinguished
Paper}{}{}{}{}

\cventry{CSL 2024~\cite{aristoteActiveLearningDeterministic2024}}{Helena Rasiowa
Award}{Best Student Paper}{}{}{}

\cventry{CSL 2024~\cite{aristoteActiveLearningDeterministic2024}}{Selected
Paper}{for publication in a special issue of the LMCS
journal~\cite{aristoteActiveLearningDeterministic2025}}{}{}{}
